Joint Field Office Situation Report 
Human Services

Date: 10/6/05

1603:
	Registrations
	1,026,685

	Total Payments
	$2,143,053,070.92

	Designated Parishes 
	31


1607:

	Registrations
	278,852

	Total Payments
	$242,073,837.39

	Designated Parishes 
	21


	DRCs 

	Fixed
	27

	Mobile
	2


	Section Reporting: 
	Other Federal Agencies

	Group Leader Contact Info:
	Clara Feldberg

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603:  

US Rural Development

Talking Points regarding their loan and grant programs for those affected by Hurricane Katrina were distributed to Other Federal Agencies Coordinators and NPSC for dissemination.

Most of the Federal Agencies have employees on site at the DRCs:

Social Security Administration, Disaster Unemployment Assistance, Internal Revenue Service, USDA-Rural Development. Other Federal Agencies continue visiting DRCs periodically, leaving brochures and phone numbers for applicants looking for Information. 

Coordinating with Office of Victims of Crime offer through one of   it’s grantees to assist FEMA in death notifications efforts to victims of           Katrina.   

Offer has been declined because FEMA does not get involved in

Death notifications

1607:  US Rural Development

Talking Points regarding their loan and grant programs for those affected by Hurricane Rita were distributed to Other Federal Agencies Coordinators and NPSC for dissemination.

 Most of the Federal Agencies have employees on site at the DRCs:

Social Security Administration, Disaster Unemployment Assistance, Internal Revenue Service, US Small Business Administration, USDA-Rural Development. Other Federal Agencies continue visiting DRCs periodically, leaving brochures and phone numbers for applicants looking for Information. 

	
	B.
	Action Items for next Operational period
	1603:        Continue coordination with OFAs in a daily basis

1607:              US POSTAL SERVICE

Post Office trailers with post office boxes inside will be located in front of the cruise ship Sensation. Keys to the PO Boxes will be issue once the trailers arrive. Residents will need to show  photo identification and their names will be checked against occupancy lists that show cabin assignments.

	
	C.
	General Status of Operations
	1603:  on target

1607: on target

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	1
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:  n/a
1607: n/a


	Section Reporting: 
	HS IA Other Programs/Special Needs Group

	Group Leader Contact Info:
	Margaret Braun 225-346-4157; 703-547-7668

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603/1607: 

· Met with representatives from the Transitional Housing Team, Groom Road site contractor, Dept of Social Services, COTAR, U.S. Dept. of Health and Human Services, and the FEMA IAO, discussed various elements of requirements for Special Needs housing; discussed current status of site operations, and availability of ADA units. Agreed that there is a need for 30 ADA units at the Groom Road site, and that placing ADA units at this site is feasible. 

· The first Special Needs Housing meeting will be next week on October 10, 2005.                      

Staff scheduled to attend: FEMA IAO, IA Housing Officer, Transitional Team Leads, Special Needs Group Lead, State IAO, a representative from the U.S. Dept. of Health and Human Services, Dept of Social Services (Housing), ERO, ESF 6 & 8, COTAR, MHOPS Contractors, and a DHS Attorney, with Office for Civil Rights and Civil Liberties. 

	
	B.
	Action Items for next Operational period
	1603/1607: 

· Get approval for placement of ADA units at Groom Road site.  

· Verify projected inventory of ADA unit types and timeframes for delivery of units to staging. 

· Identify the percentage of required ADA units for future group sites.

	
	C.
	General Status of Operations
	· In coordination with CR, PPI, and MHOPS, this week 3 special needs families were housed in travel trailers on private sites (TTPs); another 20 PPIs for special needs (TTPs) trailers are in process. Members of the 3 families called their respective SN Team representative and thanked them for their help with getting a trailer. 
· Identified unit type, (20) Full ADA Model, (10) “Accessible” Model, needed at Groom Road site.

1603:
Special Needs (SN): Referrals Cumulative: 367; Pending: 27.
Crisis Counseling (CC): CC ISP (for declared parishes) was approved with exceptions; $5,186,692.00 will be obligated; this is half of the total $10,373,384.00 award. The State will be required to submit documentation of expenditures when requesting the remaining funds. The State’s original request was for $11,490,778.00 however, this amount was reduced by the expedited $1,000,000.00 previously awarded to DMH in September. It has been recommended that the State respond by October 17, 2005 to the exceptions (five categories of conditions), prior to awarding any additional funds.

Disaster Unemployment Assistance (DUA): USDOL continues to review the LA DUA program.  There are potential discrepancies in the 9/30/05 expenditure report.

Disaster Legal Services (DLS: Continuing efforts to obtain Activity Reports.
Other Needs Assistance (ONA): No funerals have been paid.
1607:

Special Needs (SN): Referral Cumulative:9; Pending: 0.
Crisis Counseling (CC): Initiation of program pending.
Disaster Unemployment Assistance (DUA): FEMA has not received a request from the state for funding for Hurricane Rita. 
Disaster Legal Services (DLS): Initiation of program pending.
Other Needs Assistance (ONA): Ongoing.

	
	D.
	Staffing Needs
	

	
	
	
	(3) Pending completion of DAE hiring process.
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	9
	(3)
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603/1607: NA


	Section Reporting: 
	Registration Sweep Teams

	Group Leader Contact Info:
	Pat Lord

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

· We have completed all of the major shelters for registration intake.

1607:  

· We have completed over 1300 registrations.

	
	B.
	Action Items for next Operational period
	1603:    Complete special requests for registration intake. 

1607:     Determine what additional shelters we need to visit.

	
	C.
	General Status of Operations
	1603:
· 1SWEEP TEAM IS OUT ON THE FIELD TODAY.

· We have 17 staff members to respond to all requests

· The State is contacting all affected Parishes to see if they need a sweep team.

· We are finding most people are registered and have program questions.

1607:

· 4 SWEEP TEAMS ARE OUT IN THE FIELD TODAY. 

· Checking ARC list to see if any shelters need our help with registrations.

	
	D.
	Staffing Needs
	

	
	
	
	
	26# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	2 FEMA Staff / 22 contract staff 
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:
1607:


	Section Reporting: 
	ESF-6 Mass Care

	Group Leader Contact Info:
	Roger Ince; Desk: 225-334-7758; Cell: 916-541-0683

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

At current time it is not possible to separate Katrina/Rita items; working on that and hope to have it available for the next reporting period.

 1. ARF: Submitting ARF for the The Salvation Army to supply water to a kitchen.

2. Miscellaneous: Staffing has been contacted and one person is arriving 10/10/05 (Howard Hinkson) to replace Baton Rouge staff. Staff for Lake Charles AFO arriving 10/7/05 (Steve Schroder). Staff for New Orleans AFO arrived and deployed to New Orleans (Constantin Cheremeteff).

1607:

	
	B.
	Action Items for next Operational period
	1603: At current time it is not possible to separate Katrina/Rita items; working on that and hope to have it available for the next reporting period.

1. Meeting with American Red Cross and governmental officials to discuss interim/long-term housing issues and transitioning from shelters.

1607:

	
	C.
	General Status of Operations
	1603: At current time it is not possible to separate Katrina/Rita items; working on that and hope to have it available for the next reporting period.

Operations are ongoing—shelters and feeding operations are normal, no problems.

1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3at JFO; 1 in NOLA
	1 (replacement staff)
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603: N/A

1607:N/A


	Section Reporting:  
	VOLAG

	Group Leader Contact Info: 
	Harry Noftsker

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

1607:

	
	B.
	Action Items for next Operational period
	1603:

1607:

	
	C.
	General Status of Operations
	1603: 

GOODWILL:

Goodwill is in the process of securing donated medication cooler packs through Medicool.  This order will be expedited as soon as possible through the Baton Rouge Goodwill.

Goodwill received a referral to assist a triple amputee obtain a power wheelchair through the Islamic Relief Team.  Contacts with a Physical Therapist and a wheelchair provider have been made to obtain a custom power chair and a lightweight travel manual chair.  

Goodwill facilitated a wheelchair provider visit to an evacuee in a church shelter, to ascertain the evacuee’s exact needs. 
AMURT:
AMURT/EL Baton Rouge Volunteer Referral: One permanent placement with Red Cross at Kenner.

AMURT/EL Baton Rouge Collaboration: Five Volunteers assisted ICNA with sorting, and were oriented by Red Cross at Lamar Dixon Shelter in Gonzales.

AMURT/EL Baton Rouge Outreach: Five volunteers provided general support at Lamar Dixon Shelter, One volunteer provided pet therapy for three children at Addis Shelter, One volunteer recorded questions for follow up on behalf of two evacuees living out of the shelter system.
A2H:
In the last 24 hours, America’s Second Harvest had dispatched 1.2 million pounds of food to LA, which will provide an additional 937,000 meals.

DONATIONS:

Islamic Relief has purchased 150 kitchen kits, with first aid kits included, which are to be delivered tomorrow.  

Shaw Group is providing 580 kitchen kits for Baker for delivery this week.

Convoy of Hope is bringing in 580 hygiene kits.

The governor’s office is working with 3 groups with the overall goal that they purchase the 70,000 kits and donate them to this relief operation.  We hope to have this finalized by week’s end.

There is also a grant of $10,000 which will be used to buy additional kits in the short term.
LANO:

LANO has been connecting translators with Vietnamese and Hispanic communities that have been affected by the hurricanes.  LANO has also been working to locate childcare facilities (ages 0-5) for the various shelters across Baton Rouge.  We have also been notifying nonprofit agencies of the proper channels to receive financial assistance.
UNITED SIKHS:

· Continued to research the effects of the hurricanes on Sikh families in the region, entering the data into a database to ascertain which families have lost what, and what their current needs are.  

· Gave blankets and pillows to a Sikh family in need.

· Purchased pillows for Allen Chapel, and on delivery, continued to assess their needs, which include a stove as replacement for one that was stolen.

· Completed stock checks.

· Helped Chauvin shelter locate cleaning supplies. What We Need

·  In need of warehouse storage space

What We Have to Offer:

· A volunteer to work on any animal/pet projects
ARC:

General Status of Red Cross Operations in Louisiana

209,304 meals this period: 7,159,472 meals to date

256,684 snacks this period: 11,426,712 snacks to date                   

99 shelters this period; 256 shelters to date.
14,370 population period; 141,018 population to date.

21,975 rooms being utilized for transition accommodations in LA.

Community Services

As of 10-05-05, more than 20 million meals have been served by the ARC in the areas affected by hurricanes Katrina and Rita

There are 18 kitchens cooking and catering in Louisiana

Individual Client Services

Client Services will soon have more than one million cases opened 

Families Re-Connecting

Calls to 1-877-LOVED1S total 188,950 with 1,880 the calls made in the past 24 hours.

ICNA Relief:
Resolved FEMA related issues for 3 families, includes erroneous duplicate application, invalid information, and travel for family reunification.

We are having great difficulties in placing 13 Canadian medical doctors, due to certification issues.

Surveying physical condition of masjids located in Orleans, Jefferson, and St. Bernard parishes, for the purposes of establishing neighborhood assistance centers.

We assisted East Baton Rouge Parish Public School System in removing donations of clothing from their warehouse.  (To be implemented in 2-weeks.)

ACS:

New Iberia

Offloaded  2 trucks, top items

            Canned Food

Toiletries

          Shipped out 69 pallets since then, top items

            Canned Food

Diapers

Personal Care Kits

                       

Donated goods needed: Sheets, pillows

Needs for the warehouse:

   2 forklifts, 10 pallet jacks and boxes for packing 

Children’s educational toys and other miscellaneous toys are available.

SOUTHERN BAPTIST:

1. Chain Saw recovery work continuing across the state. 
2.   Beginning Blue roof project.

3. Feeding to continue with 21 Kitchens working with the Red Cross and The Salvation Army. 
4.  Reviewing request for mudout units in the New Orleans area.

5. Anticipate feeding and recovery operations to continue min. 30 -45 days.

Scientology Disaster Response:
Spiritual assists to those in need
89
Cumulative14,240

People helped
as needed 755 Cumulative
76,895

Booklets total as of today
583





Dianetics sessions cumulative  253

 We received calls of help from several small churches; one wanted shelving and the other VMs to go to help clean  elderly persons homes. The other sites are still active and going well.

We are setting up Dianetics seminars in Monroe and also Shreveport.  We are working as fast as possible to get these people Dianetics sessions so that they can then help out in rebuilding their homes.  New Orleans asked for VM help with counseling for people who lost their pets. 

We are preparing for the Dianetics workshop of the sheriffs dept and a space to hold it has been secured (Oakwood Mall on the Westbank).  

TSA:

Food Served

 

10/5/2005

Running Total

Prepared Meals (Hot or Cold)

35,885

979,798

Sandwiches

 

1,438

139,193

Cold Drinks (Soda/Juice/Water)

49,760

884,943

Snacks (Donuts, Cakes, Chips)

31,478

606,238

Other 

 

0

34,496

UNITED WAY OF ACADIANA:

The VAL/Center/United Way of Acadiana is forming a volunteer center for unaffiliated volunteers coming into the Vermillion Parish. It will be coordinated through Vermillion Faith Community of Care.

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	9
	1 to check in on 10/11/05
	6

	2.
	CURRENT ISSUES OR PROBLEMS: 

1603:

1607:


	Section Reporting: 
	ESF-11

	Group Leader Contact Info:
	Mike McIlwain

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

Disaster Mass Care Feeding (FNS)  

Food: 
State DA Orders Since Last Report
Nothing new reported from the previous report

State DA Shipments to Disaster Organizations Since Last Report
 Nothing new reported from the previous report

In response to Hurricane Katrina, Louisiana has issued 332,400 EBT cards to victims for an issuance of $206,063,069. 

Louisiana was approved to add Allen, Sabine, St. Landry, and Vernon Parishes to its waiver, and to extend the application periods for Jefferson, Orleans, Plaquemines and St. Bernard Parishes through the end of the day October 14.  Louisiana was also approved to add Evangeline parish to the DFSP.  The application period begins October 6 and expires at the end of the day October 14.
1607:

In response to Hurricane Rita, Louisiana has issued 89,768 EBT cards to victims for an issuance of $37,842,638.

On Wednesday, October 5, regional teams visited 12 FSP issuance sites, shelters and Special Supplemental Nutrition Program for Women, Infants and Children (WIC) sites in Louisiana.


	
	B.
	Action Items for next Operational period
	1603: Continue Disaster Food Stamp Program for affected parishes

1607: Continue Disaster Food Stamp Program for affected parishes

	
	C.
	General Status of Operations
	1603: Ongoing

1607: Ongoing

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	1
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 

1603: Nothing significant reported October 5, 2005

1607: Nothing significant reported October 5, 2005


	Section Reporting: 
	Housing

	Group Leader Contact Info:
	Juan Gil/Maggie D.

	1.
	SITUATION:

	
	A.
	General Status of Operations
	· Work on establishment call centers processing is proceeding in conjunction w/Dewberry and Davis.  SOW (Statement of Work) is being finalized.

· Internal processing for expedited PPI (Pre-placement Interview) processing is being refined.  PPI’s forwarded from Parishes & DRC’s (Disaster Recovery Centers) are being processed for Travel Trailers and Mobile Home request.

· Groom’s Road- Contract lessors are in the field leasing-in applicants.

·  Emergency Group Sites are being reviewed.

	
	B.
	Major Priorities Accomplished
	

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff en route
	# of Staff still needed

	
	
	
	
	
	0
	30

	2.
	CURRENT ISSUES OR PROBLEMS:  

· PPI is still in need of 20 additional Applicant Assistants.

· DHOPs is in need of 10 people to do Quality Assurance.  DHOPS will lose 23 Firemen on 10/7/05 that were requested back to their home destinations...


	Section Reporting:  
	U.S. SBA

	Group Leader Contact Info:
	Carl N. Gaspari

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: Customer Service Representatives are assigned to all DRCs to answer questions and assist individuals and business owners complete their SBA Application.  

A Personnelist is now on site to begin hiring local staff.

Identified location for a Business Recovery Center in NO

1607:

	
	B.
	Action Items for next Operational period
	1603: 

Identified location for a Business Recovery Center in NO.
1607:

	
	C.
	General Status of Operations
	1603: 

SBA continues to search out conventional lodging for members or our staff. 

Local hiring of CSRs and administrative staff is ongoing. 

Set up the Business Recovery Center in New Orleans.
Continue training of local hires in Baton Rouge.
1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	LA 220  TX 76
	5
	228

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:  Locating lodging to support staff in close proximity to the DRCs and JFO.  

1607:


	Section Reporting: 
	Barbara Copeland/Kent Sheets

	Group Leader Contact Info:
	Inspection Service Coordinators

	1.
	SITUATION:

	
	A.
	Action Items Completed 
	1603: 

There are 1216 inspectors performing inspections. 

1607: 
There are 279 inspectors performing inspections.

	
	B.
	Action Items for next Operational period
	1603:
1607:

	
	C.
	General Status of Operations
	1603:

1607:

	
	D.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	2
	0
	0

	2.
	CURRENT ISSUES OR PROBLEMS: 
1603:
1607:


	Section Reporting: 
	NPSC Liaison

	Group Leader Contact Info: 
	Pamela Glasschroeder 703-785-2661

	1.
	SITUATION:

	
	A.
	General Status of Operations
	1603/1607: 

· Reviewing Special Needs, Congressional and escalated cases for varies entities with in the JFO

· Monitoring reports and queues

· Relating information between the NPSCs and JFO 

	
	B.
	Major Priorities Accomplished
	1603/1607:
· Identified issue with applicants who should have received rental assistance due to inaccessibility however due to an issue with NEMIS received IID letters instead. Verified this was an issue which was corrected on 9/23/05

· Identified issue with postal zip code set up for Monroe Shelter applicants. Since it’s not in the postal database, NEMIS won’t accept it.  Work Around :If you change the apps in Web App Update, It should stay because address correction is disabled there.  
· Worked with representative from IGs office to provide information on how Expedited Assistance and Expedited Transitional Housing is awarded.
· Identified applicants registered under DR 1603 who should be registered under DR 1607, sent list to NPSC for transfer.
· Working on Hot Topic list for DRCs including information on ETH and family reunification


	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff pending check-in
	# of Staff still needed

	
	
	
	
	3
	0
	1

	2.
	CURRENT ISSUES OR PROBLEMS: 

· Number of UI coming from DRC, PPI, Special needs and other areas is increasing daily. Staff is attempting unsuccessfully to keep up with demand. 

· COMPUTER ISSUES continue to hamper ability to perform job functions. Slowness of wireless connection is causing liaison staff to lock applicant files delaying resolution to concern. 


	Section Reporting: 
	DRC OPS

	Group Leader and Contact Info:
	Don Baggett

	1.
	SITUATION:

	
	A.
	General Status of Operations
	DRC at Belle Chasse will be open daily from 8am – 6pm.  All other DRC’s will be open daily from 9am – 7pm.

 Fixed DRC sites now in 27 locations:

Caddo(Shreveport)

Ouacita(Monroe)

Rapides(Alexandria)

Lafayette(Lafayette)

Iberville(Plaquemine)

Plaquemines (Belle Chase)

Webster (Minden)

E. Baton Rouge (Baton Rouge)

Washington(Bogalusa)

Avoyelles(Marksville)

Washington (Franklinton)

St Tammany(Slidell)

Orleans(Algiers)

St James (Vacherie)

LaFourche(Thibodeaux)

St Charles (Boutte)

St John (LaPlace)

Ascension(Gonzales)

Jefferson(Gretna)

Iberia(New Iberia)

LaFourche(Raceland)

Tangipahoa(Amite)

Livingston (Denham Springs)

Terrebonne(Houma)

Point Coupee (New Roads)

St Tammany(Covington)

Vermillion(Abbeville)

MDRC’s:

Cruise Ship at Port of New Orleans

Washington Parish

We have also secured leases on properties at the following locations and will announce openings ASAP(In order of priority):

Opening Sunday 10/9/05:

Jefferson Davis(Jennings)  

Tentatively Sunday 10/9/05:

W. Baton Rouge(Port Allen)  

St Helena(Greensburg)  

Tentatively Monday 10/10/05:

Orleans(New Orleans#1)

Cameron(Grand Lake) ASAP

Calcasieu (Lake Charles) ASAP

St Bernard(Chalmette) ASAP

Concordia(Vidalia) ASAP

Beauregard(DeRidder) ASAP

Lincoln(Ruston) ASAP

Natchitoches(Natchitoches) ASAP

Acadia(Rayne) ASAP

Morehouse(Bastrop) ASAP

St Martin(Breaux Bridge) ASAP

St Landry(Opelousas) ASAP

Red River(Coushatta) ASAP

Franklin(Winnsboro) ASAP

Caldwell(Columbia) ASAP

Tensas(St Joseph) ASAP

Jackson(Hodge) ASAP

Vernon(Leesville) ASAP

E Feliciana(Clinton) ASAP

Winn(Winnfield) ASAP

#1 Priority: To secure inspected, approved, and leased sites at the following Parishes:  

Orleans (New Orleans#2)

Orleans (New Orleans#3)

Sabine (Many)

LaSalle(Jena)

Evangeline(Ville Platte)

Richland(Rayville)

More locations possibly to follow.

	
	B.
	Major Priorities Accomplished
	

	
	C.
	Staffing Needs
	

	
	
	
	
	# of Staff on Location
	# of Staff en route
 
	# of Staff still needed

	
	
	
	
	488
	?
	600

	2.
	CURRENT ISSUES OR PROBLEMS:   

Approximately 50% of the OFA and Firefighter personnel currently deployed to DRC’s will be reaching the end of their tour of duty by 10/07/05 and will have to be replaced.  We are using various resources to address this problem, including additional hiring under the IA TAC, and working with Admin to obtain Local Hires.

DFTO & IA Trainers are now on-site at the JFO to train all DRC personnel.
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1

